\begin{thebibliography}{10}

\bibitem{mil_train1}
Smith, R.:
\newblock Essential {T}echinques for {M}ilitary {M}odeling and {S}imulation.
\newblock Volume~1 of Simulation Conference Proceedings. (1998)  805--812

\bibitem{mil_train2}
Walton, G.H., Patton, R.M., Parsons, D.J.:
\newblock Usage {T}esting of {M}ilitary {S}imulation {S}ystems.
\newblock Winter Simulation Conference (2001)  771--779

\bibitem{mil_train3}
Service, D.M.F.P.:
\newblock Former {P}{O}{W} learns value of military training.
\newblock TRADOC News Service (2003)

\bibitem{mil_train4}
Leu, M.C., Hilgers, M.G., Agarwal, S., Hall, R.H., Lambert, T., Albright, R.,
  Nebel, K.:
\newblock Training in {V}irtual {E}nvironments for {F}irst {R}esponders.
\newblock ASEE Midwest Section Meeting (2003)

\bibitem{dem_def}
geo Community:
\newblock {U}{S}{G}{S} {S}{D}{T}{S} format {D}igital {E}levation {M}odel data
  ({D}{E}{M}).
\newblock \url{http://www.dofactory.com/Patterns/Patterns.aspx} (Accessed June
  13. 2005)

\bibitem{gis_layer_def}
Meuser, M.R.:
\newblock What is {G}{I}{S} and {H}ow {D}oes it {W}ork?
\newblock \url{http://www.dofactory.com/Patterns/Patterns.aspx} (Accessed June
  13. 2005)

\bibitem{arcgis}
ESRI:
\newblock Arc{G}{I}{S}.
\newblock \url{http://www.esri.com/software/arcgis/} (Accessed June 13. 2005)

\bibitem{bil_format}
Softwright:
\newblock Topo{S}cript .{B}{I}{L} {F}ile {F}ormat.
\newblock
  \url{http://www.softwright.com/faq/support/toposcript_bil_file_format.html}
  (Accessed April 2. 2005)

\bibitem{shp_format}
E{S}{R}{I}:
\newblock {E}{S}{R}{I} {S}hapefile {T}echnical {D}escription.
\newblock \url{http://www.esri.com/library/whitepapers/pdfs/shapefile.pdf}
  (Accessed March 17. 2005)

\bibitem{opengl}
Board, O.A.R.:
\newblock Open{G}{L} - {T}he {I}ndustry {S}tandard for {H}igh {P}erformance
  {G}raphics.
\newblock \url{http://www.opengl.org/} (Accessed June 13. 2005)

\bibitem{direct3d}
{C}orporation, M.:
\newblock Microsoft {D}irect{X}: {H}ome {P}age.
\newblock \url{http://www.microsoft.com/windows/directx/default.aspx} (Accessed
  June 13. 2005)

\bibitem{tri_dec1}
Schroeder, W.J., Zarge, J.A., Lorensen, W.E.:
\newblock Decimation of {T}riangle {M}eshes.
\newblock ACM SIGGRAPH (1992)  65--70

\bibitem{tri_dec2}
Schroeder, W.J.:
\newblock A {T}opology {M}odifying {P}rogressive {D}ecimation {A}lgorithm.
\newblock IEEE Visualization (1997)  205--212

\bibitem{mesh_dec1}
Kobbelt, L., Campagna, S., Seidel, H.:
\newblock A {G}eneral {F}ramework for {M}esh {D}ecimation.
\newblock Graphics {I}nterface {C}onference (1998)

\bibitem{mesh_dec2}
Schaufler, G., Sturzlinger, W.:
\newblock Generating {M}ultiple {L}evels of {D}etail from {P}olygonal
  {G}eometry {M}odels.
\newblock Virtual Environments \'95 (1995)  33--41

\bibitem{mesh_dec3}
Algorri, M.E., Frances-Schmitt:
\newblock Mesh {S}implification.
\newblock Volume~15 of Computer Graphics Forum. (1996) ~77

\bibitem{ear_clip_and_trap}
O'Rourke, J.:
\newblock Computational {G}eometry in {C}. 2 edn.
\newblock Cambridge {U}niversity {P}ress (1998)

\bibitem{mon_decomp}
de~Berg, M., van Kreveld, M., Overmars, M., Shcwarzkopf, O.:
\newblock Computational {G}eometry - {A}lgorithms and {A}pplications. 2 edn.
\newblock Springer (1998)

\end{thebibliography}

  The primary elements of visualization dealt with by this paper are three
  dimensional digital elevation maps (DEMs), stored using the BIL File
  Format~\cite{bil_format}, and two dimensional Geographical Information Systems
  (GIS) layers, stored using the ESRI{\texttrademark} Shapefile
  Format~\cite{shp_format}. It is common for both DEMs and GIS layers to be complex
  in terms of size and structure, thus requiring special methods for
  visualization.
  
  \subsection{Digital Elevation Map Manipulation}\label{subsec:dem_manip}
  
  Three-dimensional rendering APIs such as OpenGL~\cite{opengl} and
  Direct3D~\cite{direct3d} expect display primitives to be represented as
  collections of triangles. DEMs are easily represented using triangles, so
  rendering a DEM can be trivial. Figure~\ref{fig:dem_small} shows a DEM and all
  of the triangles used to approximate it. However, the size of modern DEMs are
  approaching the point that real-time rendering with modern graphics hardware
  is not possible. Also, this paper uses DEMs to assign an extra dimension to a
  GIS layer, not just for visualization.

  In order to efficiently deal with DEMs, this paper applies a lossy compression 
  technique to DEMs called Triangle   Decimation~\cite{tri_dec1,tri_dec2} and 
  Mesh Simplification~\cite{mesh_dec1,mesh_dec2,mesh_dec3}. Triangle 
  Decimation works by examining coincident triangles, and removing unnecessary 
  triangles. Figure~\ref{fig:tri_dec} shows a set of triangles that can  be 
  reduced and the possible effects of triangle decimation. Triangle decimation 
  is called lossy because once it is performed, it is not always possible to 
  regain the full representation of the DEM. However, the effects of the 
  decimation aim to   have little-to-no visual side effects. 
  Figure~\ref{fig:dem_small} shows a DEM that is a prime candidate for triangle 
  decimation since it has many coincident triangles that are nearly coplanar.
  
  \begin{figure}[!h]
    \begin{center}
      \includegraphics[width=3.75in]{images/tri_dec.eps}
    \end{center}
    \caption{Candidate triangles and the result of their decimation.  The small amount
             of height lost in the left area is barely noticeable in the right area.\label{fig:tri_dec}}
  \end{figure}
  
  \begin{figure}[!h]
    \begin{center}
      \includegraphics[width=3.0in]{images/approxDEMTrianglesShadow.eps}
    \end{center}
    \caption{Approximated Digital Elevation Map\label{fig:dem_small}}
  \end{figure}
  
  \subsection{GIS Layer Manipulation}\label{subsec:gis_manip}
  
  As mentioned previously, rendering APIs expect primitives to be broken into
  triangles. The GIS layers found in Shapefiles are represented using two-dimensional
  polygons. These polygons are clean, meaning they contain no
  self-intersecting lines. Examples for both clean and non-clean polygons are
  shown in Figure~\ref{fig:clean_and_unclean_poly}. Since there is no
  restriction on the   structure of the polygons beyond that of ``cleanness,''
  one cannot assume   anything of the convexity of the polygons in GIS layers.
  This paper uses   several techniques to triangulate the polygons in the GIS
  layer; ear-clipping~\cite{ear_clip_and_trap}, monotone
  decomposition~\cite{mon_decomp},   and
  trapezoidilization~\cite{ear_clip_and_trap}. The output of all these
  techniques is a   set of two-dimensional, non-intersecting triangles.
  
  \begin{figure}[!h]
    \begin{center}
      \includegraphics[width=5.0in]{images/clean_and_unclean_poly.eps}
    \end{center}
    \caption{Two clean (left, middle) polygons and a non-clean (right)
             polygon.\label{fig:clean_and_unclean_poly}}
  \end{figure}
  
  \subsection{Combining DEMs and GIS Layers}
  
  In order to combine the DEM and the GIS layer, a six step approach
  is proposed:
  \begin{enumerate}
    {\item Read and store the raw DEM and GIS layer }
    {\item Convert the GIS layer into sets of two-dimensional triangles }
    {\item Convert the DEM into triangles }
    {\item Perform triangle decimation on the DEM }
    {\item Add a third dimension to the GIS layer }
    {\item Add the DEM points to the GIS layer }
  \end{enumerate}
  
  Step 1 of the approach is trivial. Steps 2, 3, and 4 can be done using the
  algorithms discussed in~\ref{subsec:dem_manip}~and~\ref{subsec:gis_manip}.
  Step 5 can be accomplished using the $O(t_d$~$*$~$v_g)$ algorithm presented
  Figure~\ref{fig:comb_alg1}, where $t_d$ is the number of triangles in the DEM
  and $v_g$ is the number of vertices in the GIS layer. Step 6 can be computed
  using the $O(v_d$~$*$~$t_g)$ algorithm shown in Figure~\ref{fig:comb_alg2},
  where $v_d$ is the number of vertices in the DEM and $t_g$ is the number of
  triangles in the GIS layer.
  
  The algorithm relies on the ability to split a triangle into one, two, or
  three triangles new triangles. The algorithm, which we call {\bf SPLIT}, takes
  as parameters a triangle and a point. {\bf SPLIT} works by examining the
  location of the point; if the point lies on a vertex of the triangle, {\bf
  SPLIT} returns the original triangle. If the point lies on the perimeter of
  the triangles, but not on a vertex of the triangle, {\bf SPLIT} returns two
  new triangles with the point being a vertex on both triangles. Finally, if
  the point resides in the interior of the triangle, {\bf SPLIT} creates three
  new triangles using the vertices of the triangle and the new point. Examples
  for the three cases of {\bf SPLIT} can be seen in Figure~\ref{fig:tri_split}.
  In the figure, the dark dot represents the second parameter to {\bf SPLIT}.
  
    \begin{figure}
        \verb|                    |$T_0 \leftarrow $ Triangles in DEM                \\
        \verb|                    |$V_0 \leftarrow $ Vertices in GIS layer           \\
        \verb|                    |for all $t_0$ $\epsilon$ $T_0$                    \\
        \verb|                      |for all $v_0$ $\epsilon$ $V_0$                  \\
        \verb|                        |if $v_0$ has only two dimensions              \\
        \verb|                          |if $v_0$ is contained by $t_0$              \\
        \verb|                            |$v_1 \leftarrow v_0$ projected onto $t_0$ \\
        \verb|                            |$V_0 \leftarrow V_0 - \{ v_0 \}$          \\
        \verb|                            |$V_0 \leftarrow V_0 \cup \{ v_1 \}$       \\
      \caption{Algorithm for assigning height values to a GIS layer\label{fig:comb_alg1}}
    \end{figure}

    \begin{figure}
        \verb|                    |$V_0 \leftarrow $ Vertices in DEM                 \\
        \verb|                    |$T_0 \leftarrow $ Triangles in GIS layer          \\
        \verb|                    |for all $v_0$ $\epsilon$ $V_0$                     \\
        \verb|                      |for all $t_0$ $\epsilon$ $T_0$                   \\
        \verb|                        |if $v_0$ is contained by $t_0$                 \\
        \verb|                          |$T_1 \leftarrow $ {\bf SPLIT }($t_0$, $v_0$)\\
        \verb|                          |$V_0 \leftarrow V_0 - \{ v_0 \}$            \\
        \verb|                          |$T_0 \leftarrow T_0 - \{ t_0 \}$            \\
        \verb|                          |$T_0 \leftarrow T_0 \cup T_1$               \\
      \caption{Algorithm for incorporating DEM points with a GIS layer\label{fig:comb_alg2}}
    \end{figure}

  \begin{figure}
    \begin{center}
      \includegraphics[width=3.25in]{images/tri_split.eps}
    \end{center}
    \caption{Cases for {\bf SPLIT}: Splitting a triangle into (from left to right), 2
             triangles, 3 triangles, and 1 triangle.\label{fig:tri_split}}
  \end{figure}
